

Global Biogeochemical Cycles[®]

RESEARCH ARTICLE

10.1029/2022GB007678

Key Points:

- Dissolved silicon concentration and yield changed over time in rivers across biomes
- Watershed biogeochemistry played an equal or greater role than river flow regimes in driving these changes
- Long-term changes in dissolved silicon showed seasonal differences in timing and magnitude across biomes

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

K. J. Jankowski, kjankowski@usgs.gov

Citation:

Jankowski, K. J., Johnson, K., Sethna, L., Julian, P., Wymore, A. S., Shogren, A. J., et al. (2023). Long-term changes in concentration and yield of riverine dissolved silicon from the poles to the tropics. *Global Biogeochemical Cycles*, *37*, e2022GB007678. https://doi. org/10.1029/2022GB007678

Received 20 DEC 2022 Accepted 16 AUG 2023

Author Contributions:

Conceptualization: Kathi Jo Jankowski, Keira Johnson, Lienne Sethna, Paul Julian, Adam S. Wymore, Arial J. Shogren, Patrick K. Thomas, Pamela L. Sullivan, Diane M. McKnight, William H. McDowell, Ruth Heindel, Jeremy B. Jones, Benjamin Abbott, Linda Deegan, Joanna C. Carey

© 2023 The Authors. This article has been contributed to by U.S. Government employees and their work is in the public domain in the USA. This is an open access article under

the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Long-Term Changes in Concentration and Yield of Riverine Dissolved Silicon From the Poles to the Tropics

Kathi Jo Jankowski¹, Keira Johnson², Lienne Sethna³, Paul Julian⁴, Adam S. Wymore⁵, Arial J. Shogren⁶, Patrick K. Thomas⁷, Pamela L. Sullivan², Diane M. McKnight⁸, William H. McDowell⁵, Ruth Heindel⁹, Jeremy B. Jones¹⁰, Wilfred Wollheim⁵, Benjamin Abbott¹¹, Linda Deegan¹², and Joanna C. Carey¹³

¹U.S. Geological Survey, Upper Midwest Environmental Sciences Center, La Crosse, WI, USA, ²College of Earth, Ocean, and Atmospheric Science, Oregon State University, Corvallis, OR, USA, ³St. Croix Watershed Research Station, Marine on St. Croix, MN, USA, ⁴The Everglades Foundation, Palmetto Bay, FL, USA, ⁵Department of Natural Resources and the Environment, University of New Hampshire, Durham, NH, USA, ⁶Department of Biological Sciences, University of Alabama, Tuscaloosa, AL, USA, ⁷Department of Aquatic Ecology, Swiss Federal Institute of Aquatic Science and Technology (EAWAG), Dübendorf, Switzerland, ⁸Department of Civil, Environmental, and Architectural Engineering, University of Colorado Boulder, Boulder, CO, USA, ⁹Environmental Studies Program, Kenyon College, Gambier, OH, USA, ¹⁰Institute of Arctic Biology, Department of Biology and Wildlife, University of Alaska Fairbanks, Fairbanks, AK, USA, ¹¹Plant & Wildlife Sciences, Brigham Young University, Provo, UT, USA, ¹²Woodwell Climate Research Center, Falmouth, MA, USA, ¹³Mathematics, Analytics, Science, and Technology Division, Babson College, Wellesley, MA, USA

Abstract Riverine exports of silicon (Si) influence global carbon cycling through the growth of marine diatoms, which account for $\sim 25\%$ of global primary production. Climate change will likely alter river Si exports in biome-specific ways due to interacting shifts in chemical weathering rates, hydrologic connectivity, and metabolic processes in aquatic and terrestrial systems. Nonetheless, factors driving long-term changes in Si exports remain unexplored at local, regional, and global scales. We evaluated how concentrations and yields of dissolved Si (DSi) changed over the last several decades of rapid climate warming using long-term data sets from 60 rivers and streams spanning the globe (e.g., Antarctic, tropical, temperate, boreal, alpine, Arctic systems). We show that widespread changes in river DSi concentration and yield have occurred, with the most substantial shifts occurring in alpine and polar regions. The magnitude and direction of trends varied within and among biomes, were most strongly associated with differences in land cover, and were often independent of changes in river discharge. These findings indicate that there are likely diverse mechanisms driving change in river Si biogeochemistry that span the land-water interface, which may include glacial melt, changes in terrestrial vegetation, and river productivity. Finally, trends were often stronger in months outside of the growing season, particularly in temperate and boreal systems, demonstrating a potentially important role of shifting seasonality for the flux of Si from rivers. Our results have implications for the timing and magnitude of silica processing in rivers and its delivery to global oceans.

Plain Language Summary Silicon (Si) is an important nutrient for algae in freshwater and marine ecosystems and plays a fundamental role in the global carbon cycle. Dissolved Si (DSi), the bio-available form, is primarily derived from rock weathering, although many biogeochemical and hydrologic processes occurring across the land-ocean continuum control the flux of Si to coastal systems. Rivers are the main supplier of Si to the global oceans, with the flux often assumed to be unperturbed by human activities. Here, we present the first examination of multi-decadal shifts in river DSi concentrations and loads, synthesizing results from 60 sites that span nine biome types. We found widespread evidence for change in river DSi concentrations and loads in recent decades, with the largest increases occurring in alpine and polar regions. Changes were associated largely with shifts in watershed processing of DSi rather than changes in river flow regimes. The timing and magnitude of change varied according to the seasonality of temperature and precipitation across biomes. These results indicate that DSi concentration and loads are shifting over decadal timescales and are senstive to changing phenology, climate, and land use across multiple biomes.



19449224, 2023, 9, Downloaded from https://agupubs

.wiley.com

Data curation: Kathi Jo Jankowski, Lienne Sethna, Paul Julian, Adam S. Wymore, Arial J. Shogren, Pamela L. Sullivan, Diane M. McKnight, William H. McDowell, Ruth Heindel, Benjamin Abbott, Joanna C. Carey

Formal analysis: Kathi Jo Jankowski, Keira Johnson, Lienne Sethna, Patrick K. Thomas

Funding acquisition: Kathi Jo Jankowski, Joanna C. Carev Methodology: Kathi Jo Jankowski Project Administration: Kathi Jo Jankowski, Joanna C. Carev Visualization: Kathi Jo Jankowski, Keira Johnson, Paul Julian, Pamela L. Sullivan Writing - original draft: Kathi Jo Jankowski, Adam S. Wymore, Arial J. Shogren, Joanna C. Carey Writing - review & editing: Kathi Jo Jankowski, Keira Johnson, Lienne Sethna, Paul Julian, Adam S. Wymore, Arial J. Shogren, Patrick K. Thomas, Pamela L. Sullivan, Diane M. McKnight, William H. McDowell, Ruth Heindel, Jeremy B. Jones, Wilfred Wollheim, Linda Deegan, Joanna C. Carey

1. Introduction

Fluvial ecosystems transport and transform material along the land-ocean continuum, including weathering products that play critical roles in biogeochemical cycles. In the case of silicon (Si), a major weathering product, the global network of streams and rivers delivers >80% of annual Si loads to marine ecosystems (Tréguer & De La Rocha, 2013). This annual flux of Si drives global carbon (C) cycling by supporting marine diatom growth. Marine diatoms account for roughly 50% of net primary productivity in the ocean (Rousseaux & Gregg, 2013) and ~40% of the C deposited in marine benthos (Buesseler, 1998; Tréguer et al., 1995, 2018). Limitation of Si in coastal and freshwater ecosystems is typically defined by its ratio with nitrogen (N) and phosphorus (P) (Si: N < 1 and Si: P < 16), which can lead to a shift in phytoplankton species assemblages where non-siliceous species tend to outcompete diatoms (Anderson et al., 2002; Conley et al., 1993; Officer & Ryther, 1980). Such shifts in the autotrophic community have repercussions up the food chain, often with deleterious effects on marine and freshwater ecosystems. Non-siliceous algae tend to be a lower quality food source, export less C to deeper waters, and are more often associated with harmful algal blooms compared to diatoms (Guo et al., 2016; Turner et al., 2008; Wurtsbaugh et al., 2019). As such, Si supplies in marine and freshwater ecosystems exert global-scale controls on aquatic ecosystem function and C cycling.

The amount of Si exported by rivers is dictated by a variety of biogeochemical and hydrologic factors (Figure 1; Cornelis et al., 2010) operating across multiple time scales ranging from 10° to 10° years. Due to the abundance of mineral Si in the lithosphere (in the form of silicon dioxide (SiO₂); Meybeck et al., 1988; West et al., 2005; White & Blum, 1995), chemical weathering is a first order control on fluxes of Si from land to sea. The mobilization of this lithogenic Si is further influenced by regional erosion rates, basin slope, streamflow, flow path distribution, and the nature of the pedogenic pool (Asano et al., 2003; Frey et al., 2007; Gaillardet et al., 1999; Hilley & Porder, 2008; Meybeck et al., 1988). More precipitation of secondary clay minerals will result in greater Si retention along the land-sea continuum, altering rates of export (Cornelis et al., 2010). Terrestrial vegetation also regulates Si availability and flux, operating as both a source and sink of Si (Alexandre et al., 1997; Carey & Fulweiler, 2012a; Conley, 2002; Cornelis et al., 2011). Vegetative uptake has been estimated at 83 Tmol Si yr⁻¹, or roughly one-third of the total Si taken up by diatoms (Carey & Fulweiler, 2012b). Similar to diatoms, terrestrial vegetation incorporates Si into tissue, forming biogenic Si (BSi). Rates of plant Si uptake vary greatly across species (Epstein, 1994, 1999), resulting in a large degree of spatial heterogeneity in the role terrestrial plant communities play in modulating Si exports resulting from differences in land cover types within and across biomes (Carey & Fulweiler, 2012b; Conley, 2002; Cornelis et al., 2010). Terrestrial vegetation can also serve as a source of Si through the production of acidic root exudates that stimulate silicate weathering and dissolution, or through release of organic compounds that mobilize mineral Si (De Tombeur et al., 2021; Drever, 1994). Moreover, rates of BSi dissolution from plant derived organic matter degradation can be orders of magnitude faster than rates of mineral silicate weathering (Cornelis et al., 2011; Fraysse et al., 2009; Van Cappellen, 2003), resulting in a large fraction of Si being cycled through plant biomass before exported to river networks (Alexandre et al., 1997; Derry et al., 2005; Pokrovsky et al., 2013).

In-stream ecological processes, such as freshwater diatom blooms, also influence Si exports by drawing down river Si concentrations (Admiraal et al., 1990; Garnier et al., 1995). Other factors, such as N and P concentrations, light availability, invertebrate grazing, animal activity, and water residence time dictate in-stream primary productivity, diatom abundance, and Si fluxes (Conley et al., 2000; Fulweiler & Nixon, 2005; Humborg et al., 2000; Savoy & Harvey, 2021; Schoelynck et al., 2019). For example, increased water residence time in river basins can lead to the drawdown of Si through facilitating greater biotic uptake or sedimentation (Carey et al., 2019). Thus, hydro-physical and biogeochemical conditions within rivers themselves will dictate the flux of Si downstream.

Multiple processes across and within river basins exert control over fluxes of Si along the land-ocean continuum. How these processes are responding to long-term anthropogenic-driven perturbations across space and time, however, remains unknown. Compared to our knowledge of other major river solutes, such as C, N, or P (Griffiths et al., 2012; Holmes et al., 2000; Howarth, 2008; Raymond & Oh, 2007; Raymond et al., 2008), we have limited understanding of how global change is altering river Si fluxes, especially at the global scale. Regionally specific field investigations point towards potentially large shifts in river Si fluxes as a result of climate change. Increases in Si fluxes have been observed in streams of the high Arctic in areas with high rates of permafrost thaw and increased active layer depth, which was related to increases in rates of chemical weathering and soil-water interactions (Carey et al., 2020; Frey & McClelland, 2009). In temperate forested ecosystems,

2 of 22



Figure 1. Conceptual figure of the terrestrial and aquatic controls on riverine dissolved silicon (DSi) concentrations and yields. Silica derived from geogenic processes occurs either through the breakdown of rock or soil at a specific site or through the input and eventual breakdown of dust. DSi can be taken up through biotic processes (e.g., vegetation or diatoms) or precipitated as amorphous silica. DSi is transported from land to streams across various hydrologic flow paths. The interaction of the processes and their control on DSi concentrations and fluxes is dependent on the type of underlying lithology, land cover, land use, and climate. Double headed arrows in the figure indicate a wide range of variability of a given process.

long-term warming experiments have resulted in faster internal Si cycling between vegetation and soil solution (Gewirtzman et al., 2019). Tighter terrestrial Si cycling thus reduces the Si available for hydrologic transport into adjacent river ecosystems and reduces net exports. Except for a handful of studies, it remains unclear how factors such as shifting land cover and climate conditions will alter Si flux from terrestrial to aquatic ecosystems across biomes over time.

To address this knowledge gap, we evaluated trends over 15–54 years in riverine dissolved Si (DSi) across 60 rivers representing distinct biomes ranging from tropical to polar latitudes. Specifically, we addressed the following questions: (a) Are river Si concentrations and yields changing over time across biomes?, (b) Are hydrometeorological, geological, or biogeochemical processes associated with changes in Si concentrations and yields across biomes? and (c) When during the year do we observe the greatest change in Si concentrations and yields within and across biomes? We hypothesized that we would observe the largest changes in streams located in the alpine and polar regions as a result of rapid warming, glacial melt and altered hydrologic cycles, and new substrate exposure via permafrost thaw. In addition, we hypothesized that seasonal changes would vary among biomes and highlight potential climate, terrestrial vegetation, or river-specific mechanisms driving change in DSi cycling. Finally, we hypothesized that shifts in the flow regime would be a major driver of changes in DSi concentration and yield.

2. Methods

2.1. Site Description and Data Synthesis

Our data set includes published or public data sets for 60 rivers from 11 long-term research areas, primarily located in North America, but including streams in the Caribbean, several large pan-Arctic rivers and small streams in eastern Antarctica (Figure 2; Tables S1 and S2 in Supporting Information S1). The streams and rivers in this compiled data set cover nine distinct biomes (Figure 2), which included Arctic tundra, alpine tundra, polar desert, the Arctic-boreal transition, temperate coniferous forest, temperate deciduous forest, temperate grassland, tropical rainforest, and tropical seasonal forest/savanna. The large Arctic-boreal transition rivers in this study were considered to represent a transitional but distinct biome type from the Arctic tundra. Although they have a large portion of their watersheds underlain by permafrost similar to the Arctic tundra stream, they drain large regions





Figure 2. (Top panel) Location of rivers included in this study. Points are colored by biome. (Bottom panel) Mean annual temperature and precipitation for each of the long-term research stations from where river data were collected. AND = HJ Andrews Experimental Forest, ARC = Toolik Lake Arctic LTER, GRO = Great Rivers Observatory, HBR = Hubbard Brook Experimental Forest, KRR = Kissimmee River, LMP = Lamprey River Hydrologic Observatory, LUQ = Luquillo Experimental Forest, MCM = McMurdo Dry Valleys LTER, NWT = Niwot Ridge LTER, Sagehen = Sagehen Experimental Forest, UMR = Upper Mississippi River System.

that encompass both boreal forest and Arctic tundra. The data sets included in our study came from streams and rivers that ranged in size from small glacial headwater streams to some of the world's largest rivers (drainage area range: $0.1-2,950,000 \text{ km}^2$; mean annual discharge range $0.003-19,501 \text{ m}^3 \text{ s}^{-1}$; Table S2 in Supporting Information S1) and have diverse land cover, land use and climate conditions (Figure 3). We subsequently refer to streams and rivers simply as "sites" to address this large range in their size.

We limited our analyses to data sets that had at least 15 years of continuous daily discharge data, were not missing more than 4 years of DSi concentration data, and included four or more observations per year that were distributed across seasons. The mean concentration record length was 26 years but ranged from 15 to 54 years except in the Arctic tundra where we included a stream with a shorter record because it was the only Arctic stream with available data (Table S2 in Supporting Information S1). Discharge records started several years prior to concentration records as required for the load modeling approach described below. Given the rarity of studies of long-term change in river Si, and that record length did not have a significant effect on the magnitude of the trend, all available data for each site were left in the analysis (Figure S1 in Supporting Information S1). These different time scales require different interpretation, so we have noted that where applicable throughout the text.

2.2. Evaluating Trends in Concentration and Flux

We used the Weighted Regression on Time, Discharge, and Season (WRTDS) model to estimate daily concentrations and fluxes over the period of record for each river (Hirsch et al., 2010) using the *Exploration and Graphics for RivEr Trends* package in R (Hirsch & De Cicco, 2015; Hirsch et al., 2018). The WRTDS model generates four values for evaluation: modeled DSi concentration and flux as well as flow-normalized concentration and flow-normalized flux. We subsequently calculated area-normalized DSi fluxes and flow-normalized fluxes (i.e., DSi yields) to enable a more direct comparison of how much DSi is exported per unit area across watersheds. Streamflow conditions can strongly influence annual mean concentration and flux; therefore, WRTDS generates "flow-normalized" concentrations and load by integrating the C-Q relationship over the observed



Figure 3. Influence of multiple watershed characteristics in differentiating sites. Principal components analysis shows the distribution of watershed characteristics that differentiated rivers included in the study. Only variables that are significantly related to one or both axes are shown. Watershed characteristics that were significantly related to the first axis (dotted line), second axis (dashed line), or both (solid line) are plotted. Points are individual rivers and ellipses encircle all rivers from a particular biome. MAP = mean annual precipitation, MAT = mean annual temperature, and P = phosphorus.

probability distribution of stream discharge over the period of record (Choquette et al., 2019). This approach removes the effect of year-to-year variations in streamflow, such as high- or low-streamflow conditions that are largely driven by variable weather and which can be particularly influential at the start or end of a record, to provide flow-normalized values that show gradual change over time (Hirsch et al., 2010; Rumsey et al., 2021). See Text S1 in Supporting Information S1 for more information on the WRTDS model and flow-normalization. When reporting overall mean DSi concentration and yield values for each site in the data set (Figure 4), we used WRTDS modeled concentration and yield rather than discrete measured observations as they better represent conditions over a full annual cycle. When evaluating long-term trends, we used flow-normalized DSi concentration and yield so that we could assess changes over time not due to year to year variability in discharge. We statistically evaluated trends in absolute and percent change in flow-normalized concentration and yield using the



Figure 4. Distribution of annual mean dissolved silicon (DSi) concentrations and yields across biomes. Bold line in the center of the box is the median value; upper and lower ends of the boxes are 75th and 25th percentiles of the interquartile range, respectively; ends of the whiskers 1.5 times above the 75th percentile or below the 25th percentile of the interquartile range. Some site-year combinations from the polar desert streams had yields >50,000 kg km⁻² yr⁻¹, but axis was shortened to better show variability across all biomes.

EGRETci package (Hirsch et al., 2015). EGRETci uses a flexible trend estimation approach that does not assume the trend to be linear and evaluates its likelihood and uncertainty through a block bootstrapping method (Hirsch et al., 2015). We normalized the trend estimates to the number of years in the period of record for each site. We report both absolute and percent change, as the latter allows us to consider the magnitude of change in concentration and yield on the same scale (Hirsch & De Cicco, 2015). We considered trends significant if they had 70% likelihood or greater (as described by Hirsch et al. (2015)). Finally, we statistically evaluated long-term trends in annual mean discharge using a Mann-Kendall non-parametric trend test (Helsel et al., 2020) and calculated Sen's slope, which is an estimate of the median slope among all pairs of sample points, using the *trend* package in R (Pohlert, 2018). Mann-Kendall trends were considered significant at a *p*-value of ≤ 0.05 .

2.3. Potential Drivers of Change

An additional aim of this synthesis was parsing the potential long-term drivers of change in DSi concentrations and exports. To facilitate intra- and inter-biome comparisons in the drivers of DSi export, we used two approaches. First, we evaluated whether there were site-specific factors, independent of biome, that were associated with changing DSi concentration and yield, including watershed land use/land cover (LULC), lithology, drainage area, average climate (mean annual precipitation and mean annual temperature), and average river nutrient concentrations (inorganic N, P; refer to Text S1 in Supporting Information S1 for description of data used in these analysis). We compared five independent multiple regression models that included predictors from each of these five categories to assess their relative explanatory power for trends in DSi concentration and yield. We removed highly correlated predictors prior to their inclusion in each model and evaluated variable inflation factor (VIF) scores (variables with VIF > 4) were removed from the models and models were re-run. Predictors were transformed to meet normality assumptions and standardized prior to analysis to facilitate comparison of regression coefficients across predictors that varied greatly in scale. We compared models using AIC_c and report the adjusted coefficient of determination (R^2) of each model.

Second, we used an adaptation of the WRTDS model to assess whether changes in long-term concentration and vield resulted from a changing streamflow regime or shifts in watershed biogeochemical processing (Choquette et al., 2019; Murphy & Sprague, 2019; Rumsey et al., 2021). WRTDS generates this information by comparing two models, one in which the distribution of annual flows (i.e., the flow regime) is allowed to change over the period of record and the other in which it was held constant. WRTDS then estimates the portion of the overall trend explained by shifts in the flow regime alone (difference between the trends generated by the two models) versus proportion of the trend explained by changes in "watershed biogeochemical processes" (model 2; Rumsey et al., 2021). For instance, changes in concentration or yield mainly driven by changes in the flow regime (i.e., "flow regime trend") reflect shifts in the hydrologic system, such as changes in the quantity and timing of surface or groundwater flow. In contrast, trends driven by changes in watershed biogeochemistry (i.e., "biogeochemistry trend") reflect changes in DSi source dynamics (e.g., weathering rates, terrestrial uptake, river uptake) or management activities (Murphy & Sprague 2019; Rumsey et al., 2021). Although not completely independent, partitioning drivers into these broad conceptual categories provided a coarse estimate of the nature of the factors in the "hydrologic and source supply system" driving the trend in DSi (Choquette et al., 2019; Murphy & Sprague, 2019). Streamflow is often considered a central driver of changes in river DSi; thus, this approach allowed us to independently assess its relative role and whether its effect was additive or opposing from changes in source dynamics.

To quantify the relative role of changes in the streamflow regime versus biogeochemical processing in affecting concentration and yield trends, we assigned sites to these three broad categories using a ratio of the flow regime and biogeochemistry trend components: biogeochemistry-dominated, flow regime-dominated, or a mixture of both. To do so, we calculated a ratio of the absolute value of the two trend components: abs(flow regime trend)/ abs(biogeochemistry trend). A ratio of 1 would mean that both components contributed equally to the trend. To evaluate which component dominated the trend, we set threshold ratios of <0.5 for trends dominated by biogeochemistry, >1.5 for trends dominated by the flow regime, and 0.5 to 1.5 as indicating a mixture of both. We also display all values on a continuous scale in Figure 7. Finally, we assessed whether the effect of the flow regime and biogeochemistry had additive or opposing effects on the overall trend using the trend direction information (e.g., positive effect of flow change but negative effect of biogeochemical change). For these analyses, we only considered sites that had significant changes in the long-term trend in concentration or yield.

2.4. Seasonality of Change

To quantify seasonal differences among sites and biomes, we used monthly mean temperature and precipitation data to calculate annual ranges and a seasonality index (Tian et al., 2016; Van Meter et al., 2020; Walsh & Lawler, 1981) (Table S3 in Supporting Information S1). More detailed methods on monthly temperature and precipitation data and calculation of the seasonality index are provided in Texts S2 and S3 of the Supporting Information S1, respectively. To evaluate how seasonal variation in the magnitude and direction of trends in concentration and yield aligned with these seasonal patterns across biomes, we estimated monthly long-term trends for each site using the non-parametric Sen's slope estimator (Helsel et al., 2020; *trend* package in R). Trends were considered significant at p < 0.05. We were not able to assess seasonal change in polar desert streams because they were only flowing December and January each year, so we excluded them from this analysis.

3. Results

The rivers in this synthesis represented a wide range of mean annual DSi concentrations and yields (Figure 4). The global mean concentration across all sites was 4.72 mg Si L⁻¹ (median concentration = 3.22 mg Si L⁻¹) and ranged from 0.47 to 15.3 mg Si L⁻¹. The highest concentrations were observed in tropical rainforest streams in Puerto Rico and a coniferous forest catchment in California (Table S4 in Supporting Information S1). The tropical rainforest streams also exhibited the greatest range in concentration in concentration. Mean annual yield across all sites was 6,136 kg Si yr⁻¹ km⁻² (median annual yield = 2,041 kg Si yr⁻¹ km⁻²) ranging from 18.0 to 47,000 kg Si yr⁻¹ km⁻². The smallest yield was observed in a small Arctic tundra stream and the largest yields were observed in tropical savanna, tropical rainforest, and small polar desert streams in Antarctica. The greatest range in yield was exhibited by the tropical savanna and polar desert streams.

3.1. Long-Term Trends in Annual Mean Discharge

Changes over time in annual mean discharge were significant at eight sites (13%), all of which increased over time (Table S5 in Supporting Information S1). Where changes in discharge occurred, they did not necessarily change uniformly across all streams within a given region or biome. Specifically, annual mean discharge significantly increased at four polar desert sites, three temperate deciduous forest sites, and one tropical rainforest site. There were marginally significant upward trends (p < 0.10) in annual mean discharge at eight additional sites, which included one additional polar desert stream, one Arctic-boreal transition, one tropical savanna, and one temperate deciduous forest site.

3.2. Long-Term Trends in Concentration and Yield

We found significant changes in DSi concentration over time in 48 of 60 sites (80%), with changes ranging from -0.05 to 0.21 mg L⁻¹ year⁻¹ (-1.5%-10.8% per year; Table 1, Figure 5). Sites where no significant change in concentration occurred were primarily in tropical rainforest streams and temperate coniferous forest streams (Figure 6). Of the sites with significant long-term changes in concentration, there was a nearly equal distribution of increases and decreases, but more sites had increasing (n = 25) than decreasing concentrations (n = 23); Figures 5 and 6). The most substantial increases in concentrations occurred at small headwater streams in polar and alpine sites, specifically the alpine and Arctic streams, which had increases ranging from 3.45% to 10.8% per year since the 1980s for alpine streams and during the early 2000s for the Arctic tundra stream (Figure 5). More modest increases in DSi concentrations were observed across a range of biomes, including temperate deciduous forest streams in the northeastern United States, tropical savanna sites, and sites in the temperate forest and grassland biomes in the upper midwestern United States. Decreases in concentration were distributed across biomes but most common among sites in the polar desert and Arctic-boreal transition biomes, with five of six Arctic-boreal rivers and four of eight polar desert streams showing significant decreases in concentration. Decreases were also common among temperate grassland sites in Midwestern US that had agricultural influence in their watersheds. Finally, although there were some general patterns in concentration change among biomes, only the alpine tundra biome had uniform trends among all sites within a single biome, indicating that trends within other biomes likely reflected complex local watershed factors in addition to long-term regional climate shifts.

Compared to concentration changes, a similar proportion of sites showed significant change in DSi yield over time (n = 49; 82% of sites). The mean change per year in yield among all rivers was 14.3 kg ha⁻¹ yr⁻¹ (range = -8

Mean and Standard Deviation of Dissolved Stiticon (DSI) Concentration and Tieta Among Att Siles (N) within a Given Biome								
Biome	N	$\begin{array}{c} \mbox{Median} \pm \mbox{SD concentration change} \\ (\mbox{mg } L^{-1}) \end{array}$	Median ± SD concentration change (%)	Median \pm SD yield change (kg km ⁻² y ⁻¹)	Median ± SD yield change (%)			
Alpine tundra	3	0.71 ± 2.8	134.0 ± 95.1	$654 \pm 4,016$	36.3 ± 58.6			
Arctic tundra	1	0.45	166.0	13.7	121.63			
Arctic-boreal transition	6	-0.26 ± 0.34	-8.4 ± 6.9	-21.2 ± 40.1	-2.9 ± 8.2			
Polar desert	8	-0.06 ± 0.11	-6.1 ± 11.8	$115 \pm 1,571$	4.3 ± 16.2			
Temperate coniferous forest	8	0.05 ± 0.2	0.57 ± 2.2	-568 ± 719.4	-5.8 ± 7.7			
Temperate deciduous forest	16	0.27 ± 0.33	7.9 ± 9.3	230 ± 206.7	16.5 ± 12.4			
Temperate grassland	8	-0.39 ± 0.72	-9.1 ± 19.2	48.5 ± 137.7	5.1 ± 10.3			
Tropical rainforest	6	-0.18 ± 0.41	-2.5 ± 3.1	$426 \pm 2,269$	1.1 ± 8.9			
Tropical savanna	4	0.06 ± 0.41	0.89 ± 17.2	490 + 248.3	6.8 + 7.6			

Table 1

Mean and Standard Deviation of Dissolved Silicon (DSi) Concentration and Yield Among All Sites (N) Within a Given Biome

Note. Reported as the absolute or percent (%) change in concentration or yield across the full period of record for each site. (mg L^{-1} = milligrams per liter; kg km⁻² y⁻¹ = kilograms of DSi per square kilometer per year).

7–330 kg DSi ha⁻¹ yr⁻¹). Most sites had increasing (n = 37; 62% of sites) rather than decreasing DSi yield (n = 12; 20% of sites). The magnitude of percent (or relative) changes in DSi yield among biomes generally mirrored patterns in concentration, with the largest changes occurring in polar and alpine sites. The greatest relative increases in yield were observed in small alpine and Arctic streams (0.80%–7.33% per year), whereas



Figure 5. Changes in dissolved silicon (DSi) concentration and yield shown as (a) percent change in concentration per year, (b) percent change in yield per year, (c) absolute concentration change per year, and (d) absolute change per year. Bars are individual sites and are colored by biome. Error bars are 95% confidence intervals of bootstrapped trend estimates.



Global Biogeochemical Cycles

10.1029/2022GB007678



Figure 6. Proportion of sites with increasing, decreasing, and no change in dissolved silicon (DSi) (a) concentration and (b) yield by biome.

two Arctic-boreal rivers and one polar desert stream displayed the largest relative decreases in yield over time (-1.5% to -0.92% per year). Sites with no significant change in yield were mainly tropical rainforest and coniferous rainforest sites, but also included some temperate and tropical grassland sites. Patterns in absolute yield change differed somewhat from relative change, however, with large absolute yield increases occurring across a broader range of biomes. The greatest absolute yield increase was again at the highest elevation alpine tundra site, but the next greatest increases were observed in tropical rainforest and tropical savanna sites and the greatest absolute decreases occurred at tropical rainforest and temperate coniferous forest sites. Tropical rainforest and temperate coniferous forest had the smallest relative changes in yield over time but had the highest DSi concentrations among sites in our study (Figure 4), indicating that small changes in concentration in these high Si, high precipitation systems have important implications for DSi yields.

Finally, several sites had diverging trends in concentration and yield. For example, in polar desert streams, nearly uniformly increasing yields were observed, despite decreasing concentrations at 50% of sites. Such a response was due to increasing discharge at most of these sites (Table S5 in Supporting Information S1), although the concentration-discharge relationship decreased for most streams over time (Figure S3 in Supporting Information S1), indicating that processes affecting concentration decreases occurred predominantly under low flow conditions (Gooseff et al., 2017; Hirsch & De Cicco, 2015). Additionally, in coniferous forest streams, yields predominantly decreased, although most sites had either no change or upward trends in concentration. Given that there was no significant change in discharge at these sites (Table S5 in Supporting Information S1), this suggests that concentrations decreased proportionally more at higher flows.

3.3. Potential Drivers of Long-Term Change

LULC had the greatest effect on DSi trends across sites for both concentration ($R^2 = 0.42$) and yield ($R^2 = 0.30$; Table 2). When evaluating the effects of specific predictors included in the LULC model, percent shrubgrass and percent open water were the most influential variables (Table S6 in Supporting Information S1), with positive and negative effects on the trends, respectively. Percent forest cover had a significant but smaller negative effect on the trends in DSi, but percent watershed impact had no effect. After LULC, lithology was the next best model for both concentration and yield but did not explain much variation in DSi trends across sites in either case ($R^2 < 0.1$; Table 2).

Table 2

Results of Multiple Regression Analyses Evaluating the Effects of Site Specific Variables on Trends in Dissolved Silicon (DSi) Concentration and Yield

Response variable	Category	Model terms	AIC _c	ΔAIC_{c}	R^2
Concentration					
	Land Use/Land Cover	Percent forest, shrubgrass, open water, impacted	223.8	0	0.42
	Lithology	Percent volcanic, sedimentary, carbonate, metamorphic, plutonic	251.2	27.4	0.08
	Geomorphology	Drainage area	251.9	28.1	0.01
	Climate	Mean annual temperature, mean annual precipitation, latitude	253.0	29.2	0.02
	Trophic status	phosphate, nitrate	254.2	30.4	-0.01
Yield					
	Land Use/Land Cover	Percent forest, shrubgrass, open water, impacted	188.8	0	0.30
	Lithology	Percent volcanic, sedimentary, carbonate, metamorphic, plutonic	206.9	18.1	0.07
	Geomorphology	Drainage area	207.3	18.5	0.00
	Trophic status	phosphate, nitrate	208.9	20.1	-0.02
	Climate	Mean annual temperature, mean annual precipitation, latitude	209.2	20.4	0.00

Note. Models for each category of variables were run individually. Model were compared via Akaike's Information Criterion (AICc) and the coefficient of determination (R^2) . Land use/land cover and lithology variables are all percent of the watershed for each site.



Figure 7. Proposed mechanisms for driving overall trends in dissolved silicon (DSi) in (a/b) concentration and (c/d) yield across all biomes and sites. The location of the points on the quadrant figures (a and c) indicates the proportional contribution of changes in watershed biogeochemical processing (position on *y*-axis) and the flow regime (position on *x*-axis) to the overall trend. Sites were assigned to categories based on the ratio of the proportional contribution of the flow regime to biogeochemistry to the overall trend. Blue color indicates a change in the flow regime > biogeochemistry (ratio > 1.5), orange indicates change in the biogeochemistry > flow regime (ratio < 0.5), and no color indicates change in the flow regime approximately equaled that of the biogeochemistry (0.5 < ratio < 1.5). Points are colored by biome, upward/downward triangles indicate the direction of the overall trend, and the size of the point indicates the percent change. Note that scales are different between *x* and *y* axes. Panels (b and d) show the proportion of rivers within each biome that fell into each of the categories. Only sites with significant overall trends in concentration or yield were included in this figure.

Results from the WRTDS model supported these findings, indicating that long-term change in DSi concentration was related more to watershed processes (changing watershed biogeochemistry) than a changing stream flow regime at most sites (Figure 7). Of sites with significant long-term changes in DSi concentration, 42 sites (88%) had a dominant effect of watershed biogeochemistry, zero sites had a dominance of a changing flow regime, and six sites (12%) had a mixture of both (Figures 7a and 7b). Only sites in temperate biomes had an influence of both biogeochemical and flow regime change on DSi concentration trends. Changing watershed biogeochemistry was the dominant component of the trend at all other sites, including all the sites in alpine and polar regions (Arctic tundra, Arctic-boreal transition, polar desert, and alpine tundra) and all tropical sites (ratio < 0.5; Figures 7a and 7d). More streams had opposing (n = 23, 55%) effects of watershed biogeochemistry and the flow regime than additive (n = 18, 43%) on the overall trend (note: percentages exclude streams where one component had zero contribution to the trend; see position of points across quadrants in Figure 7a). Of sites where overall concentration increased, all had a positive effect of watershed biogeochemical change on the trend, but 44% had a negative effect of a change in the flow regime, indicating the negative effect of the flow regime did not exceed the positive effect of changing watershed biogeochemistry. Similarly, at sites where overall concentration decreased, all had negative effects of watershed biogeochemical change. However, a larger percentage of these sites showed an opposing effect of the flow regime, with 74% of sites exhibiting a positive effect of the flow regime on the trend despite an overall decrease in concentration. Taken together, these results indicate that changing DSi concentrations were more sensitive to shifts in watershed biogeochemical and source dynamics than shifts in the hydrologic system alone. In other words, changes in hydrology were not able to overcome the influence of changing watershed biogeochemistry on long-term concentration changes.

For DSi yields, there was a more equal distribution between long-term changes that were driven by watershed biogeochemistry versus a changing flow regime (Figures 7b and 7d). Overall, changing watershed biogeochemistry drove DSi yield changes at 21 sites (43%), changing flow regime at 23 sites (47%), and a combination of watershed biogeochemistry and the flow regime resulted in DSi yield changes at 4 sites (10%; Figure 7d). Watershed biogeochemical change dominated DSi yield trends at alpine tundra, Arctic tundra, and Arctic-boreal transition sites, none of which had any influence of a changing flow regime. In contrast, flow regime change was the dominant component of DSi yield trends at most temperate coniferous and deciduous forest sites; all but five of the temperate sites in this study had either a dominant or combined effect of a changing flow regime on changes in DSi yield. In other biomes, the drivers of change were not as consistent among sites. Polar desert streams had a nearly equal mixture of sites where changes in DSi yields were dominated by watershed biogeochemistry (n = 3) or flow regime (n = 4) and tropical rainforest and savanna sites had sites in all three categories (changing yields due to biogeochemistry, flow regime, mixture). Finally, similarly to concentration, there was a nearly equal distribution of streams that had additive (n = 23; 53%) versus opposing (n = 20; 47%) effects of changing watershed biogeochemistry and flow regime on the overall DSi yield trend (note: percentages exclude streams where one component had zero contribution to the trend; see position of points across quadrants in Figure 7c). In contrast to concentration, additive or opposing effects were also nearly equally distributed among sites that had overall upward or downward trends.

3.4. Seasonal Trends in DSi by Biome

Seasonal signals were strong in both DSi concentration and yield changes over time, but the timing and magnitude varied among biomes (Figure 8). We observed three general patterns for monthly trends in DSi concentration—a shift in the direction of changes within the year (e.g., concentration increasing over time early in the year and decreasing over time late in the year), magnitude of change greatest during coldest months of the year, and magnitude of change greatest during the warmest-wettest months (see Figure S2 in Supporting Information S1 for absolute values of temperature and precipitation). First, sites where the direction of the trend in concentration changed during the year were generally in the warmest and wettest biomes, where seasonal temperature variation was smallest but wet and dry seasons were notable: the tropical rainforest, tropical savanna, and temperate coniferous forest biomes (Figure S2, Table S3 in Supporting Information S1). Tropical rainforest stream concentrations increased during the drier season (January–June) but decreased or remained stable during the wetter season (hurricane season; July–December). Tropical savanna sites showed an inverse pattern, with decreasing concentrations during the drier portions of the year (October–February/March) but increasing during the wetter months (April–September). Temperate coniferous forest sites had the strongest decreases at the end of the wet season, but then showed either upward or no trends into the dry and early wet season (June–December).



Global Biogeochemical Cycles



Figure 8. Monthly trends in dissolved silicon (DSi) concentration (panels a, c, e, g, i, k, m, and o) and DSi yield (panels b, d, f, h, j, l, n, and p) summarized across all rivers within each biome. Trends are reported as Sen slope values (left *y*-axis). Positive values signify increasing and negative values signify decreasing over time. Polar desert excluded because long-term record only included sampling in December and January. Temperature (blue lines) and precipitation (green lines) data are normalized to their long-term mean values (right *y*-axis) and are included to represent timing of intra-annual changes. Note differences in scales across panels.

12 of 22

Sites that had the greatest magnitude change in DSi concentration during the coldest months of the year were in temperate and boreal forest regions, including Arctic-boreal transition, temperate deciduous forest, and temperate grassland sites (Figure 8; Figure S3 in Supporting Information S1). Arctic-boreal transition rivers showed consistently downward trends in DSi concentration throughout the year, but concentration decreases were largest from December to April. Temperate deciduous forest sites had increasing concentrations from November to February but had little change during the rest of the year. Temperate grassland rivers had consistent decreases observed from December to May but more variable patterns from June to November.

Finally, sites with the greatest magnitude of change during the late summer months occurred in the coldest biomes where streams were frozen in the winter and seasonal temperature variation was large– the alpine and Arctic tundra (Figures 2 and 8). In both the alpine and Arctic tundra sites, the timing of the peak change in concentration occurred on average in August–September (Figure 8). The timing varied with elevation among alpine tundra streams, however, with the greatest change in concentration occurring in May–June in the lowest elevation stream, July–September in the mid-elevation stream, and August–September in the highest elevation stream (Figure S4 in Supporting Information S1).

Seasonal DSi concentration and yield trends were not always the same within a given biome, and concentration trends often had more distinct seasonal patterns than yield trends (Figure 8). Only tropical savanna sites had a consistent seasonal shift in the direction of DSi yield trends because tropical rainforest sites had high variability in DSi yield trends during the year. There were fewer sites where the change in DSi yield was greatest during colder months of the year, and temperate coniferous sites were the only group showing this pattern. In addition, some biomes had different seasonal patterns for DSi yield versus concentration. Specifically, temperate deciduous forest sites had increasing yields across all months of the year, and trends in yield in temperate grassland sites were highly variable in magnitude and direction during the year. In Arctic-boreal transition rivers, the timing of greatest changes in yield shifted later in the year than for concentration, occurring primarily in May–June.

4. Discussion

We found widespread evidence for changes in riverine DSi concentration and yield across multiple biomes over recent decades. The most substantial shifts have occurred in some of the most rapidly warming parts of the globe-the poles and the alpine zone-but also were prevalent across sites in the temperate zone, indicating that DSi fluxes are changing on human timescales, especially in response to biogeochemical and hydrologic processes (Conley et al., 2008; Sullivan et al., 2022; Vitousek et al., 1997; Zalasiewicz et al., 2019). Changes occurred across a wide diversity of biomes, but not always in the way we hypothesized. LULC explained more variation in trends than lithology, climate, and trophic status. In addition, shifts in the flow regime were not the dominant component of long-term changes in DSi concentration or yield. This indicates that shifts in watershed biogeochemical processes are important components of change in the delivery of river Si to receiving waters (Carey & Fulweiler, 2013; Conley et al., 2008; McDowell et al., 2019). Changes in DSi were not uniform through the year, with distinct seasonal patterns to changes across biomes. For example, shifts during the coldest months of the year were notable in both temperate and boreal regions, where winters are warming rapidly (USGRCP, 2017). This demonstrates the importance of considering how differences in seasonal warming trends and changes in seasonality affect the flux of Si from rivers. Taken together, our analysis indicates that long-term shifts in river DSi are driven by diverse mechanisms, vary across biomes, and tend to be specific to different seasons across biomes.

4.1. Mean Concentrations and Yield Are Highest at Low Latitudes

Our results align with global patterns showing that DSi concentrations and yields tend to be inversely related to latitude (Beusen et al., 2009; Cornelis et al., 2010; Dürr et al., 2011; Turner et al., 2003). We observed the highest mean concentrations and yields in low latitude tropical rainforest and savanna rivers and the lowest in high latitude polar regions (Arctic tundra and polar desert streams). High temperature and precipitation in tropical systems increase weathering rates and runoff, resulting in some of the highest yields globally. Although the mean concentration in streams in our study (4.72 mg L⁻¹) was in line with the global average (4.42 mg Si L⁻¹; Dürr et al., 2011; Tréguer & De La Rocha, 2013), our range of concentrations (0.47–15.3 mg L⁻¹) was wider than other global assessments (1.8–10.9 mg L⁻¹; Dürr et al., 2011). Concentrations and yields we report here from tropical

rainforest streams are on the very high end of global ranges (Beusen et al., 2009). The tropical rainforest streams in this study were all located in Puerto Rico, a region that is classified as "hyper-active," having 5–10 times the global Si yield average (Dürr et al., 2011) and includes some of the fastest weathering siliceous terrain in the world (quartz-rich igneous bedrock; McDowell & Asbury, 1994; White & Blum, 1995). Some of these tropical streams drain steep mountainous catchments with volcanic lithology, both factors that are typically related to high Si yields (Beusen et al., 2009; Cornelis et al., 2010; McDowell et al., 1995). Temperate coniferous forest streams in our study had the next highest mean concentrations and yields. These rivers are underlain by volcanic bedrock (volcanic mudflow, andesite lava flow; Swanson & James, 1975) located on the Pacific Northwest coast of the United States., which has high yields due to the elevated tectonic activity and runoff observed in catchments that line the "Ring of Fire" (Dürr et al., 2011).

4.2. Widespread Changes in DSi Concentration and Yield, but Greatest in Alpine and Polar Regions

4.2.1. Evidence for Change

Most rivers in this study showed evidence of changing Si concentrations and yields over time. The largest changes occurred in alpine and polar regions, however, where sites showed both the largest relative increases and decreases over time. Specifically, the largest relative increases in concentration were observed in high elevation alpine tundra streams and a small stream in the Arctic tundra (Table 2, Figure 5). This aligns with previous work showing rapid change in permafrost and glacially impacted systems to be associated with increases in fluvial DSi. As high altitude and high elevation systems warm and permafrost thaws, new soil surfaces, especially where deeper mineral soils are exposed to weathering, soil-water interaction times increase, and flow paths lengthen and deepen, resulting in pulses of DSi to the fluvial network (Carey et al., 2020; Frey & McClelland, 2009; Shogren et al., 2019). The high elevation streams at Niwot Ridge in Colorado had increasing concentrations and fluxes that are likely connected to changing climate conditions (Crawford et al., 2019; Jones et al., 2012). These upward DSi trends also align with previous work linking warmer temperatures and mineral weathering to increasing sulfate and cation concentrations in these same streams (Caine, 2010; Crawford et al., 2019; Williams et al., 2006). The relatively lower rate of biomass growth and Si uptake by tundra vegetation compared to forests (Carey et al., 2017) could also allow most "new" DSi to be mobilized to streams in these systems rather than being taken up by plants.

We were only able to include a single Arctic tundra stream that had sufficient DSi and discharge data to estimate a trend in concentration and yield, but only reflected a short period (~8 years). Increased concentrations at that stream mirror those observed in other Arctic tundra streams over the same time period from the same northern Alaska research site (Figure S5 in Supporting Information S1). This recent increase (since ~2000) in DSi concentrations in small streams in this region of the Arctic could be related to climate warming, as permafrost thaw increases in northern Alaska (Bowden et al., 2008; Caine, 2010; Schuur et al., 2022) could increase stream DSi (Frey & McClelland, 2009). However, decadal climate oscillations (e.g., Pacific Decadal Oscillation) could be an additional driver of these observations (Jones et al., 2012), as other streams at the same research area with longer Si records had decreasing observed concentrations over the 10 year period prior to 2000 (Figure S6 in Supporting Information S1; data from these streams was not included in analysis because of a lack of corresponding discharge data). Whether our single Arctic tundra stream over this period reflects common trends in DSi across a very diverse Arctic biome is not well understood. Continuing and expanding long-term monitoring and analysis of both chemistry and hydrology in streams across the high Arctic would be beneficial for better understanding how this stream fits into broader patterns across a rapidly changing region of the globe.

Increases in DSi concentration and yield were also widespread across temperate sites, specifically sites in the temperate deciduous forest biome, where 81% and 94% of sites had increasing DSi concentration and yield, respectively. The increases in concentration and yield are meaningful when considered in absolute terms, with the period of record concentration changes ranging from -0.66 to 0.85 mg L⁻¹ and yield ranging from -146 to 764 kg km⁻² yr⁻¹. These increases were consistent across the geographic regions of the temperate forest sites in our data set (i.e., the Midwestern and Northeastern United States), despite large variation in their historical or current land use (Data Set S1; Figures S7–S9 in Supporting Information S1) and differences in time scales (Table 1). Increasing DSi concentration and yields occurred across sites with and without historical deforestation and sites with a high proportion of cropland in their watersheds (Figures S7 and S9 in Supporting Information S1). For example, several sites in the Midwestern United States had agricultural land use in their watersheds,

but trend directions and magnitudes were independent of land use across rivers (Figure S9 in Supporting Information S1), which is similar to previous work showing that DSi concentration and yields in this region were not strongly related to land use (Carey et al., 2019). The consistently increasing river DSi concentrations and yields across the diverse temperate forest biome indicate that shifts in climate, weathering, forest composition, or recovery from acid deposition (Likens et al., 1996; Rodríguez-Cardona et al., 2022) may overwhelm current or historical LULC-driven differences at these sites. More work exploring the role of land use and agriculture on DSi dynamics would be useful (Sethna et al., 2022).

Although most sites had increasing DSi yields, some deviations from that pattern were notable in large rivers draining to the Arctic Ocean where we observed widespread decreases in DSi concentration and yield since the early 2000s. Several of those basins have a large percentage of permafrost cover in their watersheds (26%-100%; Holmes et al., 2012). Given previous work showing increasing DSi flux over time in permafrost-dominated systems (Frey & McClelland, 2009) and positive relationships between active layer depth and DSi loads (Carey et al., 2020), we expected DSi increases in these rivers similar to what we observed in small alpine and Arctic streams. Several factors could be driving declines in DSi that differ from drivers in small streams. This region of the globe is warming rapidly, which can increase the productivity of both the terrestrial and aquatic ecosystems (Myers-Smith et al., 2020; Song et al., 2018), thereby increasing DSi uptake on land and in the rivers. We found a negative correlation between basin forest cover and river DSi exports (Table 2), similar to previous work at lower latitudes (Carey & Fulweiler, 2013; Chen et al., 2014). Thus, given the large stocks of boreal forest biomass in their basins (Data Set S1), the terrestrial Si "sink" is potentially substantial in these watersheds and may also explain the divergence of trends between these rivers and increases in Arctic and alpine tundra streams. Furthermore, warming temperatures have the potential to increase aquatic productivity (Song et al., 2018) and reduce the duration of river ice cover in rivers around the globe (Yang et al., 2020). Thus, warmer temperature, shorter duration or thinner ice cover, and greater N and P mobilization from thawing permafrost may result in greater primary productivity (and Si uptake) in the rivers during more of the year (Jankowski et al., 2021). Finally, other work has shown multi-decadal declines in sediment export from many of these rivers because of sedimentation in upstream reservoirs (Zolkos et al., 2022). Particulate materials are often rich in Si and correlated with river DSi (Durr et al., 2011); therefore, if upstream sediment retention remains high, from historic dam construction for example, it may counteract any recent increases in Si derived from thawing permafrost or deepening active layers (Frings et al., 2014; Humborg et al., 2000; Maavara et al., 2014).

Although further work would be beneficial to understand the mechanisms driving these unexpected downward trends, such decreases have potentially substantial implications for phytoplankton productivity in the downstream coastal and marine environment of the Arctic Ocean (Baines et al., 2012; Carey et al., 2020; Pabi et al., 2008) as these large Arctic rivers account for over half of the discharge entering the Arctic basin. The declines represent a relative change in DSi yield of -17.8%-1.9% and an absolute change of -91.5-8.9 kg DSi km⁻² yr⁻¹ across the pan-Arctic region. These changes are small, but not insignificant relative to the mean annual loads of DSi of Arctic rivers (323–1,370.5 kg km⁻² yr⁻¹; Carey et al., 2020; Dürr et al., 2011; Holmes et al., 2012) and could have implications for marine productivity in the Arctic Ocean if the trends continue.

4.2.2. Mixed Results

Despite uniform increases in concentrations and yields in small Arctic and alpine sites and the widespread decreases across large Arctic-boreal transition rivers, other biomes revealed mixed results among streams within them, including polar desert, tropical savanna, tropical rainforest, and temperate grassland streams. Rivers within the temperate grassland biome had a particularly wide range of concentration changes, from -26% to 25%. These differences could result from several watershed-specific factors, such as differences in watershed lithology, soil composition and properties, and biological processing (e.g., plant or algal uptake), which become more important in determining biogeochemical patterns at more local scales (Rodríguez-Cardona et al., 2022). Watershed lithology and soil composition affect both the amount of Si available for dissolution, as well as the water residence time and flow paths that transport Si from landscape to river.

In addition, several biomes had differences in the magnitude or direction of concentration versus yield trends. For example, polar desert streams had either decreasing or no change in DSi concentrations but nearly uniformly increasing DSi yields. Observed decreases in concentration in the polar desert may be due to several factors. There have been shifts in diatom biomass over time in some streams, although diatom communities in some streams have shifted from highly silicified to small-bodied diatoms with lower demand for Si (Stanish et al., 2011).

Alternatively, as a major source of Si to these streams comes from hyporheic weathering (Hirst et al., 2020), decreasing concentrations in these streams could instead be from the persistent depletion of hyporheic Si reservoirs due to consistently higher discharge depleting readily dissolvable secondary clay minerals and amorphous Si from hyporheic zones (Hatton et al., 2020; Hirst et al., 2020). Thus, pulses of DSi at high flows may occur when the stream is connected to residual Si source areas, but ongoing depletion from higher median flows could result in differences in the direction of change in flow-normalized concentration and yield (Wlostowski et al., 2016).

Some sites showed small relative changes over time but large absolute magnitude of change, such as tropical rainforest and temperate coniferous rainforest sites (Figure 5), which drain watersheds with the highest terrestrial biomass. The temperate coniferous forest sites included in this study are located in the Pacific Northwest region of the United States, and despite evidence for increases in temperature (Abatzoglou et al., 2014; Jones et al., 2012), forests in both of these regions have shown little biological response to documented climate change in more recent time periods (Cowles et al., 2021; Jones et al., 2012). In tropical rainforest streams, the absolute magnitude of change ranged from 62% to 110% of the mean annual yield from tropical rivers (4,064 kg DSi km⁻² yr⁻¹). These large absolute changes can be explained by the combination of high DSi concentrations (Figure 4) and precipitation, as well as their location in "highly active" areas of the Earth for DSi (Durr et al., 2011). If these trends are widespread, small relative changes in flux are associated with large absolute change in flux from these watersheds can have a disproportionate influence on marine processes locally and globally.

4.3. Land Cover and Watershed Biogeochemistry, Not Flow Regime, Associated With Most Long-Term Changes in River DSi

DSi flux from terrestrial to aquatic ecosystems depends on the relative contribution of different pools of Si (e.g., mineralogical and biological) as well as watershed hydrology driving subsurface and surface water flow (Beusen et al., 2009; Struyf et al., 2009), all factors that vary across the diverse biomes included in this study. Accordingly, we found that differences in LULC across sites were more strongly associated with differences in long-term trends in DSi concentration and yield than lithology, climate, drainage area, or N or P availability (Table 2). In addition, although river discharge is often considered a primary driver of DSi concentration and flux (Beusen et al., 2009; Turner et al., 2003), our results demonstrated a greater importance of changes in watershed biogeochemical cycling, compared to the flow regime alone, in fueling observed changes in DSi over time (Figure 7). Our results indicate that changes in the availability and mobility of DSi in the stream-watershed were more influential on long-term changes in concentrations than changes in the amount and timing of water moving through the hydrologic system. "Watershed biogeochemistry" captures several potential processes that span the terrestrial-aquatic continuum (Figure 1), including terrestrial Si cycling and uptake, river productivity or primary producer communities, weathering rates, or altered ground or surface water flow paths that affect the connections with Si sources and sinks. These categories of watershed biogeochemistry and flow regime are not independent of course, as biogeochemical shifts could be linked to changes in the flow regime if higher magnitude flows connect rivers with new sources of Si or changes in flood timing affect the capacity for uptake in the river (i.e., earlier floods, cooler temperatures, lower algal uptake). However, our finding highlights that factors driving changes in DSi concentration and yield go far beyond simple changes in discharge or flow regime (Murphy & Sprague, 2019).

The shifting biogeochemical signal was particularly strong at polar and alpine sites (Arctic tundra, alpine tundra, Arctic-boreal transition), where most trends were dominated by biogeochemical changes for both concentration and yield. These systems are experiencing rapid rates of change with respect to permafrost thaw, glacial melt, and increasing air temperatures, all of which may alter biogeochemical processes without changing annual mean discharge or other aspects of the flow regime. The exception to this pattern was the polar desert streams, where changing flow regimes had a greater role in explaining yield trends over time across sites. Most polar desert streams had significant or marginally significant (p < 0.10; Table S5 in Supporting Information S1) shifts in annual mean discharge over the period of record, consistent with other studies (Gooseff et al., 2017). Thus, even though concentrations decreased over time in most polar desert streams, the simultaneous shift in the flow regime drove increases in yield. These disparate patterns in trends and dominant mechanisms of change across alpine, Arctic, boreal and polar desert streams highlight the heterogeneity of polar and alpine regions, emphasizing that greater understanding of biogeochemical shifts occurring in these rapidly changing parts of the biosphere would be beneficial.

As expected, DSi vield was more evenly influenced by biogeochemistry and streamflow changes compared to concentration. Temperate sites had the most evidence for the role of shifting flow regime in driving changes in DSi yield, either as a mixed effect of biogeochemistry and flow (21% and 11% of temperate sites for concentration and yield, respectively) or as dominance of the flow regime alone (0% and 65% of temperate sites for concentration and yield, respectively). Shifts in flow regime encapsulate changes in timing, duration, frequency, and magnitude of flows (Poff et al., 1997), all of which likely contributed to the changes in DSi that we observed (Webster et al., 2016). For example, increases in river flow over the last several decades have occurred across multiple seasons in the Upper Mississippi River Basin (Van Appledorn, 2022) where several of the sites that had DSi trends dominated by a shift in the flow regime were located. Across all biomes, changes in annual mean discharge were significant at more than 50% of the sites where a changing flow regime dominated the yield trend. We did not explicitly quantify other aspects of a changing flow regime that could have influenced DSi yield, such as changes in minimum or maximum flows, changes in timing, or changes in flood frequency (Poff et al., 1997), but our findings indicate that some other aspects of flow regime shifts were important at sites where annual mean discharge did not change. Finally, we found a nearly equal mixture of additive and opposing effects of the flow regime and biogeochemical change on trends in DSi. This indicates that the relative role and direction of biogeochemical and hydrologic processes can vary, and better understanding of their relative contributions to Si dynamics will help us understand and forecast how Si yields might change in the future.

4.4. Seasonal Timing of Change Shows Some Common Patterns Across Biomes

Climate-driven seasonal shifts in hydrology and temperature have implications for the timing of nutrient loads (Seybold et al., 2022; Shogren et al., 2020), which can affect local and downstream productivity and food web dynamics (Myrstener et al., 2021; Woodward et al., 2010). We observed distinct seasonal patterns to long-term changes in DSi concentrations, and to a lesser degree in DSi yields.

Trend directions differed during the year in the warmest and wettest biomes (i.e., tropical rainforest, tropical savanna, and temperate coniferous forests), which had relatively stable temperatures but variable seasonality in precipitation. These sites also showed the smallest relative change in DSi over time (Figure 5), perhaps because of these intra-annual differences in the direction of change. The seasonal differences in the direction of change over time corresponded to seasonal precipitation, such as the timing of either the wet/dry season (temperate coniferous forest, tropical savanna) or the timing of the hurricane season (tropical rainforest; McDowell et al., 2013). Changes in yield were much more variable throughout the year than concentration, however. Precipitation, and therefore discharge, can be stochastic at these sites (McDowell et al., 2012) and many tropical rainforest ecosystems. Thus, it follows that seasonal changes in DSi yield would not necessarily mirror changes in DSi concentration and may be more difficult to predict into the future (Dalling et al., 2016; Zimmerman et al., 2021).

Biomes where the greatest change occurred during the coldest parts of the year included temperate deciduous forest, temperate grassland, and Arctic-boreal transition biomes, where the magnitude of monthly trends tended to be inversely related to temperature (Figure 8). Increasing concentrations during winter temperate deciduous forest systems may be a lagged signal of increases in weathering rates earlier in the year because weathering products may be stored in the groundwater and transported to the river during high flow periods. In contrast, the winter decline in Arctic-Boreal rivers a could reflect a lagged signal of greater terrestrial uptake during the summer that is detectable only in winter months (Crossman et al., 2016; Shousha et al., 2021). Cold season changes in DSi in large Arctic-boreal watersheds could also reflect earlier ice-out and spring freshets in these systems (Bintanja & Andry, 2017; Box et al., 2019; Feng et al., 2021), which may drive greater productivity in late winter or early spring periods that draw down river DSi.

The timing of change was not always consistent between concentration and yield within a given biome. Concentrations tended to have more clear seasonal patterns than yield, which may indicate that changing concentrations are more tightly linked to seasonally variable controls such as aquatic and terrestrial productivity. In contrast, yield is typically tightly coupled to changes in discharge (Carey & Fulweiler, 2013; Fulweiler & Nixon, 2005), which could be tied to a different suite of drivers or may not be changing as consistently during the year as biological processes in these regions.

These seasonal patterns indicate that considering only trends in the long-term annual average may miss important temporal shifts in the timing of peak DSi concentration and yield. The timing of change not only provides information regarding the mechanism driving the long-term shift in DSi but also has important ecological and biogeochemical implications for rivers and downstream oceans. Our finding of greater change in concentrations during the fall or winter in temperate and boreal systems has different implications for rivers and oceans than similar changes during the summer months. Increasing availability of DSi during colder times of the year could affect the development of riverine algal communities and river productivity. For example, studies in lake ecosystems have shown linkages between fall availability of DSi and the subsequent spring diatom bloom (Katz et al., 2015). Similarly, increasing higher yields to the ocean during colder months of the year may have implications for diatom uptake, annual plankton community development, and resulting carbon burial rates. Future work could focus on understanding the mechanisms driving these changes and whether the patterns observed in some months and rivers will persist or shift as watersheds respond to ongoing warming and changing hydrology.

4.5. Conclusions and Next Steps

Si is vastly understudied in rivers relative to other nutrients, despite its ecological importance for freshwater and marine ecosystems. Here, we present the first known analyses of long-term trends in DSi in rivers and streams across multiple biomes and ecosystem types. We show widespread evidence for dynamic changes in river DSi over time, with the largest shifts occurring in the streams and rivers of polar and alpine regions. In addition, we show that many changes are occurring outside of the growing season, indicating that climate-driven shifts in seasonality could affect the seasonal cycle of Si concentrations and export. Finally, using a trend attribution approach, we were able to separate the effect of changes in the flow regime from changes in watershed biogeochemistry on long-term trends in DSi, showing that changes in streamflow regimes were not the dominant driver of long-term changes in DSi. These findings emphasize the dynamic nature of river DSi and the potential for large and meaningful changes as ecosystems warm. Our work focused on quantifying the total change in Si over time using records that spanned various time scales, but further work could examine dynamics (e.g., threshold vs. linear change over time) and mechanisms driving the more recent shifts we observed, particularly in high latitude systems where changes may be rapid, abrupt, and ongoing. Future work could also investigate mechanisms driving the observed changes, potentially through examining other Si fractions in export water (e.g., BSi) and through the use of geochemical (i.e., germanium/Si ratios) or isotopic tracers. Finally, ongoing and increased monitoring of riverine Si would be useful for filling out our understanding of the Si cycle and how it may change over the long term. More long-term monitoring would be especially useful in areas outside the temperate zone and in months outside the growing season, where fewer data are available to understand the implications for concentrations and fluxes of river Si.

Data Availability Statement

The data supporting this manuscript can be accessed at https://doi.org/10.5066/P951UKQB (Jankowski et al., 2023). Citations for original data sources are given in Table S1 of the Supporting Information S1.

References

Abatzoglou, J. T., Rupp, D. E., & Mote, P. W. (2014). Questionable evidence of natural warming of the northwestern United States. *Proceedings* of the National Academy of Sciences of the United States of America, 111(52). https://doi.org/10.1073/pnas.1421311112

Admiraal, W., Breugem, P., Jacobs, D. M. L. H. A., & De Ruyter Van Steveninck, E. D. (1990). Fixation of dissolved silicate and sedimentation of biogenic silicate in the lower river Rhine during diatom blooms. *Biogeochemistry*, 9(2), 175–185. https://doi.org/10.1007/BF00692170 Alexandre, A., Meunier, J.-D., Colin, F., & Koud, J.-M. (1997). Plant impact on the biogeochemical cycle of silicon and related weathering

processes. Geochimica et Cosmochimica Acta, 61(3), 677–682. https://doi.org/10.1016/S0016-7037(97)00001-X

Anderson, D. M., Glibert, P. M., & Burkholder, J. M. (2002). Harmful algal blooms and eutrophication: Nutrient sources, composition, and consequences. *Estuaries*, 25(4), 704–726. https://doi.org/10.1007/BF02804901

Asano, Y., Uchida, T., & Ohte, N. (2003). Hydrologic and geochemical influences on the dissolved silica concentration in natural water in a steep headwater catchment. *Geochimica et Cosmochimica Acta*, 67(11), 1973–1989. https://doi.org/10.1016/S0016-7037(02)01342-X

Baines, S. B., Twining, B. S., Brzezinski, M. A., Krause, J. W., Vogt, S., Assael, D., & McDaniel, H. (2012). Significant silicon accumulation by marine picocyanobacteria. *Nature Geoscience*, 5(12), 886–891. https://doi.org/10.1038/ngeo1641

Beusen, A. H. W., Bouwman, A. F., Dürr, H. H., Dekkers, A. L. M., & Hartmann, J. (2009). Global patterns of dissolved silica export to the coastal zone: Results from a spatially explicit global model. *Global Biogeochemical Cycles*, 23(4), 1–13. https://doi.org/10.1029/2008GB003281

Bintanja, R., & Andry, O. (2017). Towards a rain-dominated Arctic. Nature Climate Change, 7(4), 263–267. https://doi.org/10.1038/nclimate3240
Bowden, W. B., Gooseff, M. N., Balser, A., Green, A., Peterson, B. J., & Bradford, J. (2008). Sediment and nutrient delivery from thermokarst features in the foothills of the North Slope, Alaska: Potential impacts on headwater stream ecosystems: Thermokarst impacts on stream ecosystems. Journal of Geophysical Research, 113(G2), G02026. https://doi.org/10.1029/2007JG000470

This work was completed as part of a synthesis working group entitled From Poles to Tropics: A multi-biome synthesis investigating the controls on river Si exports and was supported through the Long Term Ecological Research Network Office (LNO) (NSF award numbers 1545288 and 1929393) and the National Center for Ecological Analysis and Synthesis, UCSB awarded to KJJ and JCC. KJJ was supported by the US Army Corps of Engineers Upper Mississippi River Restoration Program. Funding was provided to J. Carey from the Babson Faculty Research Fund and the Andrew J. Butler and Debi Butler Term Chair. Funding was provided to K. Johnson and P. Sullivan by the Department of Energy Grant DE-SC0020146; and from National Science Foundation 2034232 and 2012796. We thank all people across the National Science Foundation Long-Term Ecological Research (NSF LTER) Network, the Upper Mississippi River Restoration Program, the Arctic Great River Observatory, and the Lamprey River Observatory who collected and analyzed samples and provided the data. We thank the University of Puerto Rico, the Luquillo LTER (DEB 1831592) Luquillo CZO (EAR 1331841), and the United States Forest Service International Institute of Tropical Forestry for assistance in collecting and processing samples from the tropical rainforest streams. The National Center for Ecological Analysis and Synthesis provided technical and administrative support and we thank Kristen Peach, Julien Brun, Nicolas Lyon, and Angel Chen for help with compiling data from the Environmental Data Initiative and generating watershed land cover and lithology data. We thank Anna Wright and Mike Gooseff in generating watershed areas for McMurdo LTER streams; Max Holmes, Lindsay Scott, and Anya Suslova for help with data and interpretion of large Arctic-boreal transition rivers trends: and Jennifer Murphy for her consultation and insight into the WRTDS model and trend attribution approaches. Finally, we are grateful to two anonymous reviewers and Becky Kreiling for their thoughtful comments and suggestions that improved the manuscript. Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government.

- Box, J. E., Colgan, W. T., Christensen, T. R., Schmidt, N. M., Lund, M., Parmentier, F.-J. W., et al. (2019). Key indicators of Arctic climate change: 1971–2017. *Environmental Research Letters*, *14*(4), 045010. https://doi.org/10.1088/1748-9326/aafc1b
- Buesseler, K. O. (1998). The decoupling of production and particulate export in the surface ocean. *Global Biogeochemical Cycles*, *12*(2), 297–310. https://doi.org/10.1029/97GB03366
- Caine, N. (2010). Recent hydrologic change in a Colorado alpine basin: An indicator of permafrost thaw? Annals of Glaciology, 51(56), 130–134. https://doi.org/10.3189/172756411795932074
- Carey, J. C., & Fulweiler, R. W. (2012a). Human activities directly alter watershed dissolved silica fluxes. *Biogeochemistry*, 111(1–3), 125–138. https://doi.org/10.1007/s10533-011-9671-2
- Carey, J. C., & Fulweiler, R. W. (2012b). The terrestrial silica pump. *PLoS One*, 7(12), e52932. https://doi.org/10.1371/journal.pone.0052932
- Carey, J. C., & Fulweiler, R. W. (2013). Watershed land use alters riverine silica cycling. *Biogeochemistry*, 113(1-3), 525–544. https://doi.org/10.1007/s10533-012-9784-2
- Carey, J. C., Gewirtzman, J., Johnston, S. E., Kurtz, A., Tang, J., Vieillard, A. M., & Spencer, R. G. M. (2020). Arctic River dissolved and biogenic silicon exports—Current conditions and future changes with warming. *Global Biogeochemical Cycles*, 34(3). https://doi. org/10.1029/2019GB006308
- Carey, J. C., Jankowski, K., Julian, P., Sethna, L. R., Thomas, P. K., & Rohweder, J. (2019). Exploring silica stoichiometry on a large floodplain riverscape. Frontiers in Ecology and Evolution, 7, 346. https://doi.org/10.3389/fevo.2019.00346
- Carey, J. C., Parker, T. C., Fetcher, N., & Tang, J. (2017). Biogenic silica accumulation varies across tussock tundra plant functional type. Functional Ecology, 31(11), 2177–2187. https://doi.org/10.1111/1365-2435.12912
- Chen, N., Wu, Y., Wu, J., Yan, X., & Hong, H. (2014). Natural and human influences on dissolved silica export from watershed to coast in Southeast China: Influences on watershed silica export. *Journal of Geophysical Research: Biogeosciences*, 119(1), 95–109. https://doi.org/10.1002/2013JG002429
- Choquette, A. F., Hirsch, R. M., Murphy, J. C., Johnson, L. T., & Confesor, R. B. (2019). Tracking changes in nutrient delivery to western Lake Erie: Approaches to compensate for variability and trends in streamflow. *Journal of Great Lakes Research*, 45(1), 21–39. https://doi. org/10.1016/j.jglr.2018.11.012
- Conley, D., Schelske, C., & Stoermer, E. (1993). Modification of the biogeochemical cycle of silica with eutrophication. Marine Ecology Progress Series, 101, 179–192. https://doi.org/10.3354/meps101179
- Conley, D. J. (2002). Terrestrial ecosystems and the global biogeochemical silica cycle. *Global Biogeochemical Cycles*, 16(4), 68-1–68–8. https://doi.org/10.1029/2002GB001894
- Conley, D. J., Likens, G. E., Buso, D. C., Saccone, L., Bailey, S. W., & Johnson, C. E. (2008). Deforestation causes increased dissolved silicate losses in the Hubbard Brook Experimental Forest: Dissolved silicate losses. *Global Change Biology*, 14(11), 2548–2554. https://doi. org/10.1111/j.1365-2486.2008.01667.x
- Conley, D. J., Stålnacke, P., Pitkänen, H., & Wilander, A. (2000). The transport and retention of dissolved silicate by rivers in Sweden and Finland. Limnology & Oceanography, 45(8), 1850–1853. https://doi.org/10.4319/lo.2000.45.8.1850
- Cornelis, J.-T., Delvaux, B., Cardinal, D., André, L., Ranger, J., & Opfergelt, S. (2010). Tracing mechanisms controlling the release of dissolved silicon in forest soil solutions using Si isotopes and Ge/Si ratios. *Geochimica et Cosmochimica Acta*, 74(14), 3913–3924. https://doi. org/10.1016/j.gca.2010.04.056
- Cornelis, J.-T., Delvaux, B., Georg, R. B., Lucas, Y., Ranger, J., & Opfergelt, S. (2011). Tracing the origin of dissolved silicon transferred from various soil-plant systems towards rivers: A review. *Biogeosciences*, 8(1), 89–112. https://doi.org/10.5194/bg-8-89-2011
- Cowles, J., Templeton, L., Battles, J. J., Edmunds, P. J., Carpenter, R. C., Carpenter, S. R., et al. (2021). Resilience: Insights from the U.S. LongTerm ecological research network. *Ecosphere*, 12(5). https://doi.org/10.1002/ecs2.3434
- Crawford, J. T., Hinckley, E.-L. S., Litaor, M. I., Brahney, J., & Neff, J. C. (2019). Evidence for accelerated weathering and sulfate export in high alpine environments. *Environmental Research Letters*, 14(12), 124092. https://doi.org/10.1088/1748-9326/ab5d9c
- Crossman, J., Catherine Eimers, M., Casson, N. J., Burns, D. A., Campbell, J. L., Likens, G. E., et al. (2016). Regional meteorological drivers and long term trends of winter-spring nitrate dynamics across watersheds in northeastern North America. *Biogeochemistry*, 130(3), 247–265. https://doi.org/10.1007/s10533-016-0255-z
- Dalling, J. W., Heineman, K., González, G., & Ostertag, R. (2016). Geographic, environmental and biotic sources of variation in the nutrient relations of tropical montane forests. *Journal of Tropical Ecology*, 32(5), 368–383. https://doi.org/10.1017/S0266467415000619
- Derry, L. A., Kurtz, A. C., Ziegler, K., & Chadwick, O. A. (2005). Biological control of terrestrial silica cycling and export fluxes to watersheds. *Nature*, 433(7027), 728–731. https://doi.org/10.1038/nature03299
- De Tombeur, F., Cornelis, J.-T., & Lambers, H. (2021). Silicon mobilisation by root-released carboxylates. Trends in Plant Science, 26(11), 1116–1125. https://doi.org/10.1016/j.tplants.2021.07.003
- Drever, J. I. (1994). The effect of land plants on weathering rates of silicate minerals. *Geochimica et Cosmochimica Acta*, 58(10), 2325–2332. https://doi.org/10.1016/0016-7037(94)90013-2
- Dürr, H. H., Meybeck, M., Hartmann, J., Laruelle, G. G., & Roubeix, V. (2011). Global spatial distribution of natural riverine silica inputs to the coastal zone. *Biogeosciences*, 8(3), 597–620. https://doi.org/10.5194/bg-8-597-2011
- Epstein, E. (1994). The anomaly of silicon in plant biology. Proceedings of the National Academy of Sciences of the United States of America, 91(1), 11–17. https://doi.org/10.1073/pnas.91.1.11
- Epstein, E. (1999). Silicon. Annual Review of Plant Physiology and Plant Molecular Biology, 50(1), 641–664. https://doi.org/10.1146/annurev. arplant.50.1.641
- Feng, D., Gleason, C. J., Lin, P., Yang, X., Pan, M., & Ishitsuka, Y. (2021). Recent changes to Arctic river discharge. Nature Communications, 12(1), 6917. https://doi.org/10.1038/s41467-021-27228-1
- Fraysse, F., Pokrovsky, O. S., Schott, J., & Meunier, J.-D. (2009). Surface chemistry and reactivity of plant phytoliths in aqueous solutions. *Chemical Geology*, 258(3–4), 197–206. https://doi.org/10.1016/j.chemgeo.2008.10.003
- Frey, K. E., & McClelland, J. W. (2009). Impacts of permafrost degradation on arctic river biogeochemistry. *Hydrological Processes*, 23(1), 169–182. https://doi.org/10.1002/hyp.7196
- Frey, K. E., Siegel, D. I., & Smith, L. C. (2007). Geochemistry of west Siberian streams and their potential response to permafrost degradation: Geochemistry of west Siberian streams. *Water Resources Research*, 43(3), W03406. https://doi.org/10.1029/2006WR004902
- Frings, P. J., De La Rocha, C., Struyf, E., van Pelt, D., Schoelynck, J., Hudson, M. M., et al. (2014). Tracing silicon cycling in the Okavango Delta, a sub-tropical flood-pulse wetland using silicon isotopes. *Geochimica et Cosmochimica Acta*, 142, 132–148. https://doi.org/10.1016/j. gca.2014.07.007
- Fulweiler, R. W., & Nixon, S. W. (2005). Terrestrial vegetation and the seasonal cycle of dissolved silica in a southern New England coastal river. *Biogeochemistry*, 74(1), 115–130. https://doi.org/10.1007/s10533-004-2947-z

- Gaillardet, J., Dupré, B., Louvat, P., & Allègre, C. J. (1999). Global silicate weathering and CO₂ consumption rates deduced from the chemistry of large rivers. *Chemical Geology*, 159(1–4), 3–30. https://doi.org/10.1016/S0009-2541(99)00031-5
- Garnier, J., Billen, G., & Coste, M. (1995). Seasonal succession of diatoms and Chlorophyceae in the drainage network of the Seine River: Observation and modeling. *Limnology & Oceanography*, 40(4), 750–765. https://doi.org/10.4319/lo.1995.40.4.0750
- Gewirtzman, J., Tang, J., Melillo, J. M., Werner, W. J., Kurtz, A. C., Fulweiler, R. W., & Carey, J. C. (2019). Soil warming accelerates biogeochemical silica cycling in a temperate forest. Frontiers in Plant Science, 10, 1097. https://doi.org/10.3389/fpls.2019.01097
- Gooseff, M. N., Barrett, J. E., Adams, B. J., Doran, P. T., Fountain, A. G., Lyons, W. B., et al. (2017). Decadal ecosystem response to an anomalous melt season in a polar desert in Antarctica. *Nature Ecology & Evolution*, 1(9), 1334–1338. https://doi.org/10.1038/s41559-017-0253-0
- Griffiths, N. A., Tank, J. L., Royer, T. V., Warrner, T. J., Frauendorf, T. C., Rosi-Marshall, E. J., & Whiles, M. R. (2012). Temporal variation in organic carbon spiraling in Midwestern agricultural streams. *Biogeochemistry*, 108(1–3), 149–169. https://doi.org/10.1007/s10533-011-9585-z
- Guo, F., Kainz, M. J., Sheldon, F., & Bunn, S. E. (2016). The importance of high-quality algal food sources in stream food webs—Current status and future perspectives. *Freshwater Biology*, 61(6), 815–831. https://doi.org/10.1111/fwb.12755
- Hatton, J. E., Hendry, K. R., Hirst, C., Opfergelt, S., Henkel, S., Silva-Busso, A., et al. (2020). Silicon isotopic composition of dry and wet-based glaciers in Antarctica. Frontiers in Earth Science, 8, 286. https://doi.org/10.3389/feart.2020.00286
- Helsel, D. R., Hirsch, R. M., Ryberg, K. M., Archfield, S. A., & Gilroy, E. J. (2020). Statistical methods in water resources. In U.S. geological survey techniques and methods (Vol. book 4, p. 458). U.S. Geological Survey.
- Hilley, G. E., & Porder, S. (2008). A framework for predicting global silicate weathering and CO₂ drawdown rates over geologic time-scales. Proceedings of the National Academy of Sciences of the United States of America, 105(44), 16855–16859. https://doi.org/10.1073/ pnas.0801462105
- Hirsch, R. M., Archfield, S. A., & De Cicco, L. A. (2015). A bootstrap method for estimating uncertainty of water quality trends. Environmental Modelling & Software, 73, 148–166. https://doi.org/10.1016/j.envsoft.2015.07.017
- Hirsch, R. M., & De Cicco, L. A. (2015). User guide to exploration and graphics for RivEr trends (EGRET) and dataRetrieval: R packages for hydrologic data (techniques and methods). U.S. Geological Survey.
- Hirsch, R. M., De Cicco, L. A., Watkins, D., Carr, L., & Murphy, J. C. (2018). EGRET: Exploration and graphics for RivEr trends (version 3.0). Hirsch, R. M., Moyer, D. L., & Archfield, S. A. (2010). Weighted regressions on time, discharge, and season (WRTDS), with an applica-
- tion to Chesapeake Bay river inputs. JAWRA Journal of the American Water Resources Association, 46(5), 857–880. https://doi. org/10.1111/j.1752-1688.2010.0048..x
- Hirst, C., Opfergelt, S., Gaspard, F., Hendry, K. R., Hatton, J. E., Welch, S., et al. (2020). Silicon isotopes reveal a non-glacial source of silicon to crescent stream, McMurdo Dry Valleys, Antarctica. Frontiers in Earth Science, 8, 229. https://doi.org/10.3389/feart.2020.00229
- Holmes, R. M., McClelland, J. W., Peterson, B. J., Tank, S. E., Bulygina, E., Eglinton, T. I., et al. (2012). Seasonal and annual fluxes of nutrients and organic matter from large rivers to the Arctic Ocean and surrounding seas. *Estuaries and Coasts*, 35(2), 369–382. https://doi.org/10.1007/ s12237-011-9386-6
- Holmes, R. M., Peterson, B. J., Gordeev, V. V., Zhulidov, A. V., Meybeck, M., Lammers, R. B., & Vörösmarty, C. J. (2000). Flux of nutrients from Russian rivers to the Arctic Ocean: Can we establish a baseline against which to judge future changes? *Water Resources Research*, 36(8), 2309–2320. https://doi.org/10.1029/2000WR900099
- Howarth, R. W. (2008). Coastal nitrogen pollution: A review of sources and trends globally and regionally. Harmful Algae, 8(1), 14-20. https://doi.org/10.1016/j.hal.2008.08.015
- Humborg, C., Conley, D. J., Rahm, L., Wulff, F., Cociasu, A., & Ittekkot, V. (2000). Silicon retention in river basins: Far-reaching effects on biogeochemistry and aquatic food webs in coastal marine environments. AMBIO: A Journal of the Human Environment, 29(1), 45–50. https:// doi.org/10.1579/0044-7447-29.1.45
- Jankowski, K. J., Carey, J. C., Julian, P., Johnson, K., Sethna, L. R., Thomas, P. K., et al. (2023). Dissolved silicon concentration and yield estimates from streams and rivers in North America and Antarctica, 1964-2021 [Dataset]. U.S. Geological Survey Data Release. https://doi. org/10.5066/P951UKQB
- Jankowski, K. J., Houser, J. N., Scheuerell, M. D., & Smits, A. P. (2021). Warmer winters increase the biomass of phytoplankton in a large Floodplain River. Journal of Geophysical Research: Biogeosciences, 126(9), e2020JG006135. https://doi.org/10.1029/2020JG006135
- Jones, J. A., Creed, I. F., Hatcher, K. L., Warren, R. J., Adams, M. B., Benson, M. H., et al. (2012). Ecosystem processes and human influences regulate streamflow response to climate change at long-term ecological research sites. *BioScience*, 62(4), 390–404. https://doi.org/10.1525/ bio.2012.62.4.10
- Katz, S. L., Izmest'eva, L. R., Hampton, S. E., Ozersky, T., Shchapov, K., Moore, M. V., et al. (2015). The "M elosira years" of Lake B aikal: Winter environmental conditions at ice onset predict under-ice algal blooms in spring. *Limnology & Oceanography*, 60(6), 1950–1964. https:// doi.org/10.1002/lno.10143
- Likens, G. E., Driscoll, C. T., & Buso, D. C. (1996). Long-term effects of acid rain: Response and recovery of a forest ecosystem. Science, 272(5259), 244–246. https://doi.org/10.1126/science.272.5259.244
- Maavara, T., Dürr, H. H., & Van Cappellen, P. (2014). Worldwide retention of nutrient silicon by river damming: From sparse data set to global estimate. Global Biogeochemical Cycles, 28(8), 842–855. https://doi.org/10.1002/2014GB004875
- McDowell, W. H., & Asbury, C. E. (1994). Export of carbon, nitrogen, and major ions from three tropical montane watersheds. *Limnology & Oceanography*, 39(1), 111–125. https://doi.org/10.4319/lo.1994.39.1.0111
- McDowell, W. H., Brereton, R. L., Scatena, F. N., Shanley, J. B., Brokaw, N. V., & Lugo, A. E. (2013). Interactions between lithology and biology drive the long-term response of stream chemistry to major hurricanes in a tropical landscape. *Biogeochemistry*, 116(1–3), 175–186. https://doi.org/10.1007/s10533-013-9916-3
- McDowell, W. H., Lugo, A. E., & James, A. (1995). Export of nutrients and major ions from Caribbean catchments. Journal of the North American Benthological Society, 14(1), 12–20. https://doi.org/10.2307/1467721
- McDowell, W. H., McDowell, W. G., Potter, J. D., & Ramírez, A. (2019). Nutrient export and elemental stoichiometry in an urban tropical river. *Ecological Applications*, 29(2). https://doi.org/10.1002/eap.1839
- McDowell, W. H., Scatena, F. N., Waide, R. B., Brokaw, N. V., Camilo, G. R., Covich, A., et al. (2012). Geographic and ecological setting of the Luquillo Mountains. In N. V. Brokaw, T. A. Crowl, A. E. Lugo, W. H. McDowell, F. N. Scatena, R. B. Waide, et al. (Eds.), A Caribbean forest tapestry: The multidimensional nature of disturbance and response. Oxford University Press.
- Meybeck, M., Cauwet, G., Dessery, S., Somville, M., Gouleau, D., & Billen, G. (1988). Nutrients (organic C, P, N, Si) in the eutrophic River Loire (France) and its estuary. *Estuarine, Coastal and Shelf Science*, 27(6), 595–624. https://doi.org/10.1016/0272-7714(88)90071-6
- Murphy, J., & Sprague, L. (2019). Water-quality trends in US rivers: Exploring effects from streamflow trends and changes in watershed management. Science of the Total Environment, 656, 645–658. https://doi.org/10.1016/j.scitotenv.2018.11.255

- Myers-Smith, I. H., Kerby, J. T., Phoenix, G. K., Bjerke, J. W., Epstein, H. E., Assmann, J. J., et al. (2020). Complexity revealed in the greening of the Arctic. *Nature Climate Change*, *10*(2), 106–117. https://doi.org/10.1038/s41558-019-0688-1
- Myrstener, M., Gómez-Gener, L., Rocher-Ros, G., Giesler, R., & Sponseller, R. A. (2021). Nutrients influence seasonal metabolic patterns and total productivity of Arctic streams. *Limnology & Oceanography*, 66(S1). https://doi.org/10.1002/lno.11614
- Officer, C., & Ryther, J. (1980). The possible importance of silicon in marine eutrophication. *Marine Ecology Progress Series*, *3*, 83–91. https://doi.org/10.3354/meps003083
- Pabi, S., van Dijken, G. L., & Arrigo, K. R. (2008). Primary production in the Arctic Ocean, 1998–2006. Journal of Geophysical Research, 113(C8), C08005. https://doi.org/10.1029/2007JC004578
- Poff, N. L., Allan, J. D., Bain, M. B., Karr, J. R., Prestegaard, K. L., Richter, B. D., et al. (1997). The natural flow regime. *BioScience*, 47(11), 769–784. https://doi.org/10.2307/1313099
- Pohlert, T. (2018). Trend: Non-parametric trend tests and change-point detection (version R package version 1(0)).
- Pokrovsky, O. S., Reynolds, B. C., Prokushkin, A. S., Schott, J., & Viers, J. (2013). Silicon isotope variations in Central Siberian rivers during basalt weathering in permafrost-dominated larch forests. *Chemical Geology*, 355, 103–116. https://doi.org/10.1016/j.chemgeo.2013.07.016
- Raymond, P. A., & Oh, N.-H. (2007). An empirical study of climatic controls on riverine C export from three major U.S. watersheds: Climate and river C export. *Global Biogeochemical Cycles*, 21(2). https://doi.org/10.1029/2006GB002783
- Raymond, P. A., Oh, N.-H., Turner, R. E., & Broussard, W. (2008). Anthropogenically enhanced fluxes of water and carbon from the Mississippi River. *Nature*, 451(7177), 449–452. https://doi.org/10.1038/nature06505
- Rodríguez-Cardona, B. M., Wymore, A. S., Argerich, A., Barnes, R. T., Bernal, S., Brookshire, E. N. J., et al. (2022). Shifting stoichiometry: Long-term trends in stream-dissolved organic matter reveal altered C:N ratios due to history of atmospheric acid deposition. *Global Change Biology*, 28(1), 98–114. https://doi.org/10.1111/gcb.15965
- Rousseaux, C., & Gregg, W. (2013). Interannual variation in phytoplankton primary production at a global scale. *Remote Sensing*, 6(1), 1–19. https://doi.org/10.3390/rs6010001
- Rumsey, C. A., Miller, O., Hirsch, R. M., Marston, T. M., & Susong, D. D. (2021). Substantial declines in salinity observed across the Upper Colorado River Basin during the 20th century, 1929–2019. Water Resources Research, 57(5), e2020WR028581. https://doi. org/10.1029/2020WR028581
- Savoy, P., & Harvey, J. W. (2021). Predicting light regime controls on primary productivity across CONUS river networks. Geophysical Research Letters, 48(10), e2020GL092149. https://doi.org/10.1029/2020GL092149
- Schoelynck, J., Subalusky, A. L., Struyf, E., Dutton, C. L., Unzué-Belmonte, D., Van de Vijver, B., et al. (2019). Hippos (*Hippopotamus amphibius*): The animal silicon pump. *Science Advances*, 5(5), eaav0395. https://doi.org/10.1126/sciadv.aav0395
- Schuur, E. A. G., Abbott, B. W., Commane, R., Ernakovich, J., Euskirchen, E., Hugelius, G., et al. (2022). Permafrost and climate change: Carbon cycle feedbacks from the warming Arctic. Annual Review of Environment and Resources, 47(1), 343–371. https://doi.org/10.1146/ annurev-environ-012220-011847
- Sethna, L. R., Royer, T. V., Speir, S., Trentman, M. T., Mahl, U. H., Hagemeier, L. P., & Tank, J. L. (2022). Silicon concentrations and stoichiometry in two agricultural watersheds: Implications for management and downstream water quality. *Biogeochemistry*, 159(2), 1–18. https:// doi.org/10.1007/s10533-022-00927-7
- Seybold, E. C., Dwivedi, R., Musselman, K. N., Kincaid, D. W., Schroth, A. W., Classen, A. T., et al. (2022). Winter runoff events pose an unquantified continental-scale risk of high wintertime nutrient export. *Environmental Research Letters*, 17(10), 104044. https://doi. org/10.1088/1748-9326/ac8be5
- Shogren, A. J., Zarnetske, J. P., Abbott, B. W., Iannucci, F., & Bowden, W. B. (2020). We cannot shrug off the shoulder seasons: Addressing knowledge and data gaps in an Arctic headwater. *Environmental Research Letters*, 15(10), 104027. https://doi.org/10.1088/1748-9326/ab9d3c
- Shogren, A. J., Zarnetske, J. P., Abbott, B. W., Iannucci, F., Frei, R. J., Griffin, N. A., & Bowden, W. B. (2019). Revealing biogeochemical signatures of Arctic landscapes with river chemistry. *Scientific Reports*, 9(1), 12894. https://doi.org/10.1038/s41598-019-49296-6
- Shousha, S., Maranger, R., & Lapierre, J. (2021). Different forms of carbon, nitrogen, and phosphorus influence ecosystem stoichiometry in a north temperate river across seasons and land uses. *Limnology & Oceanography*, 66(12), 4285–4298. https://doi.org/10.1002/lno.11960
- Song, C., Dodds, W. K., Rüegg, J., Argerich, A., Baker, C. L., Bowden, W. B., et al. (2018). Continental-scale decrease in net primary productivity in streams due to climate warming. *Nature Geoscience*, 11(6), 415–420. https://doi.org/10.1038/s41561-018-0125-5
- Stanish, L. F., Nemergut, D. R., & McKnight, D. M. (2011). Hydrologic processes influence diatom community composition in Dry Valley streams. Journal of the North American Benthological Society, 30(4), 1057–1073. https://doi.org/10.1899/11-008.1
- Struyf, E., Smis, A., Van Damme, S., Meire, P., & Conley, D. J. (2009). The global biogeochemical silicon cycle. Silicon, 1(4), 207–213. https:// doi.org/10.1007/s12633-010-9035-x
- Sullivan, P. L., Billings, S. A., Hirmas, D., Li, L., Zhang, X., Ziegler, S., et al. (2022). Embracing the dynamic nature of soil structure: A paradigm illuminating the role of life in critical zones of the Anthropocene. *Earth-Science Reviews*, 225, 103873. https://doi.org/10.1016/j. earscirev.2021.103873
- Swanson, F. J., & James, M. E. (1975). Geology and geomorphology of the HJ Andrews experimental forest, western Cascades, Oregon. U.S. Department of Agriculture, Forest Service. Pacific Northwest Forest and Range Experiment Station.
- Tian, S., Youssef, M. A., Richards, R. P., Liu, J., Baker, D. B., & Liu, Y. (2016). Different seasonality of nitrate export from an agricultural watershed and an urbanized watershed in Midwestern USA. *Journal of Hydrology*, 541(Part B), 1375–1384. https://doi.org/10.1016/j. jhydrol.2016.08.042
- Tréguer, P., Bowler, C., Moriceau, B., Dutkiewicz, S., Gehlen, M., Aumont, O., et al. (2018). Influence of diatom diversity on the ocean biological carbon pump. *Nature Geoscience*, 11(1), 27–37. https://doi.org/10.1038/s41561-017-0028-x
- Tréguer, P., Nelson, D. M., Van Bennekom, A. J., DeMaster, D. J., Leynaert, A., & Queguiner, B. (1995). The silica balance in the world ocean: A reestimate. *Science*, 268(5209), 375–379. https://doi.org/10.1126/science.268.5209.375

Tréguer, P. J., & De La Rocha, C. L. (2013). The world ocean silica cycle. Annual Review of Marine Science, 5(1), 477–501. https://doi.org/10.1146/annurev-marine-121211-172346

Turner, R. E., Rabalais, N. N., & Justic, D. (2008). Gulf of Mexico hypoxia: Alternate states and a legacy. *Environmental Science & Technology*, 42(7), 2323–2327. https://doi.org/10.1021/es071617k

Turner, R. E., Rabalais, N. N., Justic', D., & Dortch, Q. (2003). Global patterns of dissolved N, P and Si in large rivers. (Vol. 22).

- USGRCP. (2017). Climate science special report: Fourth national climate assessment (Vol. 1). United States Global Change Research Program. https://doi.org/10.7930/J0J964J6
- Van Appledorn, M. (2022). Hydrologic indicators (p. 18). U.S. Geological Survey.
- Van Cappellen, P. (2003). Biomineralization and global biogeochemical cycles. Reviews in Mineralogy and Geochemistry, 54(1), 357–381. https://doi.org/10.2113/0540357

- Van Meter, K. J., Chowdhury, S., Byrnes, D. K., & Basu, N. B. (2020). Biogeochemical asynchrony: Ecosystem drivers of seasonal concentration regimes across the Great Lakes Basin. *Limnology & Oceanography*, 65(4), 848–862. https://doi.org/10.1002/lno.11353
- Vitousek, P. M., Mooney, H. A., Lubchenco, J., & Melillo, J. M. (1997). Human domination of Earth's ecosystems. *Science*, 277, 7. Walsh, R. P. D., & Lawler, D. M. (1981). Rainfall seasonality: Description, spatial patterns, and change through time. *Weather*, *36*(7), 201–208.
- https://doi.org/10.1002/j.1477-8696.1981.tb05400.x
 Webster, J. R., Knoepp, J. D., Swank, W. T., & Miniat, C. F. (2016). Evidence for a regime shift in nitrogen export from a forested watershed. *Ecosystems*, 19(5), 881–895. https://doi.org/10.1007/s10021-016-9974-1
- West, A., Galy, A., & Bickle, M. (2005). Tectonic and climatic controls on silicate weathering. Earth and Planetary Science Letters, 235(1–2), 211–228. https://doi.org/10.1016/j.epsl.2005.03.020
- White, A. F., & Blum, A. E. (1995). Effects of climate on chemical, weathering in watersheds. *Geochimica et Cosmochimica Acta*, 59(9), 1729–1747. https://doi.org/10.1016/0016-7037(95)00078-e
- Williams, M. W., Knauf, M., Caine, N., Liu, F., & Verplanck, P. L. (2006). Geochemistry and source waters of rock glacier outflow, Colorado Front Range. Permafrost and Periglacial Processes, 17(1), 13–33. https://doi.org/10.1002/ppp.535
- Wlostowski, A. N., Gooseff, M. N., McKnight, D. M., Jaros, C., & Lyons, W. B. (2016). Patterns of hydrologic connectivity in the McMurdo Dry Valleys, Antarctica: A synthesis of 20 years of hydrologic data. *Hydrological Processes*, 30(17), 2958–2975. https://doi.org/10.1002/ hyp.10818
- Woodward, G., Perkins, D. M., & Brown, L. E. (2010). Climate change and freshwater ecosystems: Impacts across multiple levels of organization. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 365(1549), 2093–2106. https://doi.org/10.1098/rstb.2010.0055
- Wurtsbaugh, W. A., Paerl, H. W., & Dodds, W. K. (2019). Nutrients, eutrophication and harmful algal blooms along the freshwater to marine continuum. Wiley Interdisciplinary Reviews: Water, 6(5). https://doi.org/10.1002/wat2.1373
- Yang, X., Pavelsky, T. M., & Allen, G. H. (2020). The past and future of global river ice. Nature, 577(7788), 69–73. https://doi.org/10.1038/ s41586-019-1848-1
- Zalasiewicz, J., Waters, C. N., Williams, M., & Summerhayes, C. P. (Eds.) (2019). The Anthropocene as a geological time unit: A guide the the scientific evidence and current debate. Cambridge University Press.
- Zimmerman, J. K., Wood, T. E., González, G., Ramirez, A., Silver, W. L., Uriarte, M., et al. (2021). Disturbance and resilience in the Luquillo experimental forest. *Biological Conservation*, 253, 108891. https://doi.org/10.1016/j.biocon.2020.108891
- Zolkos, S., Zhulidov, A. V., Gurtovaya, T. Y., Gordeev, V. V., Berdnikov, S., Pavlova, N., et al. (2022). Multidecadal declines in particulate mercury and sediment export from Russian rivers in the pan-Arctic basin. *Proceedings of the National Academy of Sciences of the United States of America*, 119(14), e2119857119. https://doi.org/10.1073/pnas.2119857119